Dear editor and reviewers:

ROUND 1
Thank you for your letter and the reviewers’ comments on our manuscript entitled "Systemic lupus erythematosus presented with progressive massive ascites and CA-125 elevation, Tjalma Syndrome? Case report and review of the literature (NO: 77283)". We thank the editor and reviewers for the appreciation of the contents of the manuscript. The comments of Reviewer #1 are very important, we have reviewed the process of this case and found the clinical characteristics of the patient do meet the manifestations of lupus cystitis. So we have added the discussion of this issue. The abstract has been rewritten focusing on the core tip of this case report. The title of this manuscript has changed into "Systemic lupus erythematosus presenting with progressive massive ascites and CA-125 elevation, Tjalma syndrome? A case report and literature review". We have supplied the latest cutting-edge case to improving the content of the manuscript and modified the format of the table and figure. Dr. Xianfeng Zhang provide the modification guide and foundation of polish and publication, so we have updated the foundation and the author list. At last, the manuscript has been polished by a professional English language editing company and I have provided the English Language Certificate issued by the company. Reviewer #2 Re-provide the manuscript. Reviewer #3 Thanks for your comment.

After being checked and amened as seen appropriate, we hope that standard of this manuscript satisfies the requirements of submission to journals to be considered for publication.

Kind regards,

Wang Jundi

E-mail: 21818187@zju.edu.cn

Corresponding author: Huang Jiao

E-mail address: huangjiao2001@163.com
ROUND 2
The revised MS is much improved. However, the style of Ref. papers are somewhat strange. This should be corrected to universal standard style.
The reference has been revised.